Author’s response to reviews

Title: Awareness, Attitudes, and Practices Related to the Swine Influenza Pandemic among the Saudi Public

Authors:

Hanan H Balkhy (Balkhyh@ngha.med.sa)
Mostafa A Abolfotouh (mabolfotouh@gmail.com)
Rawabi H Al-Hathlool (r.alhathlool@gmail.com)
Mohammed A Al-Jumah (Jumahm@ngha.med.sa)

Version: 3 Date: 11 February 2010

Author's response to reviews:

De Editor,

Many thanks for the review comments you sent for revision. The manuscript was modified according to these further comments by reviewers. The following was done:
1. All spelling mistakes and grammatical errors were corrected.
2. The word “Panic” was changed in all the text of the manuscript as recommended by the reviewer.

All modifications and or corrections are highlighted in red.

I hope the manuscript in its final version will be suitable for publication in the Biomed Infectious Diseases Journal.

Sincerely yours,

Mostafa Abolfotouh, MD, DrPH, FASHA